Author’s response to reviews

Title: The burden of prolonged indwelling catheter after acute urinary retention in Ikeja - Lagos, Nigeria

Authors:

   Stephen O Ikuerowo (kerowq@yahoo.com)
   A A Ogunade (forever@yahoo.com)
   Taiwo O Ogunlowo (drogunlowo@yahoo.co.uk)
   Charles C Uzodinma (parkason@doctor.com)
   Julius O Esho (olusanmiesho@yahoo.com)

Version: 5 Date: 29 August 2007

Author’s response to reviews:

Dear Editor,

We have made the following changes:

1. We have included a statement indicating that informed consent was obtained from the patients for the study.
2. The abstract has been formatted to <350 words

We hope we have provided answers to all the comments.

Thank you and with kind regards.